Author’s response to reviews

Title: Assessment of need for Inpatient Treatment for Mental Disorder among Female Prisoners: A Cross-Sectional Study of Provincially Detained Women in Ontario

Authors:

Roland Jones (roland.jones@camh.ca)

Kiran Patel (kiran.patel@camh.ca)

Alexander Simpson (sandy.simpson@camh.ca)

Version: 1 Date: 08 Oct 2018

Author’s response to reviews:

Thank you for your comments.

1. The ethics statement includes the name of the institution that approved it and the reason why consent was not required
2. A conclusions section has been added to the manuscript